
JUC, 08-06-2018, 11.00

present Harro, Des, Jonathan, Benito, Arpad

Des: fixed script, Jonathan confirms it is fixed. However, data has
disappeared, ask Bob!

Jonathan: will investigate weights of 0.5

Benito: wil start looking at extended source next week, after EVN
session done

Next e on 19th of June. Will check next week what Fila firmware is on,
ask them to change to new if they want. Need to know what stations,
actually by now we do... Then do test during test period before
observations.


